Author’s response to reviews

Title: Hearing the voices of older adult patients: processes and findings to inform health services research

Authors:

Sally Fowler Davis (s.fowler-davis@shu.ac.uk)
Anne Silvester (anne.silvester@sth.nhs.uk)
Deborah Barnett (deborah.barnett@sth.nhs.uk)
Lisa Farndon (Lisa.Farndon@sth.nhs.uk)
Mubarak Ismail (M.M.Ismail@shu.ac.uk)

Version: 2 Date: 28 Nov 2018

Author’s response to reviews:

Thank you for identifying the errors in referencing accuracy. The references have now be thoroughly edited in accordance with the journal's house style. All references have been retained but some are not linked.

I hope this is satisfactory and you will kindly accept this draft for peer review

Best wishes

Sally

Dr Sally Fowler Davis